Author's response to reviews

Title: Missed Opportunities for Tobacco Use Screening and Brief Cessation Advice in South African Primary Health Care: a Cross-sectional Study

Authors:

olufemi B Omole (alagbaomole@gmail.com)
Kabilabe NW Ngobale (kabilabemrn@telkomsa.net)
Olalekan A Ayo-Yusuf (lekan.ayo-yusuf@up.ac.za)

Version: 4 Date: 24 November 2010

Author's response to reviews:

Dear Editorial team,

The following is a list of the changes made to the revised version:

1. A functional web link created for reference 15
2. The authors’ contribution constructed into a paragraph.

Thanks

Dr. OB Omole